Author’s response to reviews

Title: Comparative actions of progesterone, medroxyprogesterone acetate, drospirenone and nestorone on breast cancer cell migration and invasion

Authors:

Fu Xiao-Dong (fuxiaodong27@hotmail.com)
Maria Silvia Giretti (mariasilviagiretti@virgilio.it)
Lorenzo Goglia (lorenzo@obgyn.med.unipi.it)
Marina Ines Flamini (flaminimarinaines@hotmail.com)
Angel Matias Sanchez (angelsanchez2001@hotmail.com)
Chiara Baldacci (drsajekil@hotmail.com)
Silvia Garibaldi (silvia.Garibaldi@obgyn.med.unipi.it)
Regine Sitruk Ware (regine@popcbr.rockefeller.edu)
Andrea R Genazzani (argenazzani@tiscali.it)
Tomasso Simoncini (t.simoncini@obgyn.med.unipi.it)

Version: 7 Date: 30 April 2008

Author’s response to reviews: see over
To Melissa Norton, MD
Editor-in-Chief
BMC Cancer

Dear Dr. Norton,

I am pleased to resubmit to BMC Cancer the revised manuscript 2939652831784705 R2:

“COMPARATIVE ACTIONS OF PROGESTERONE, MEDROXYPROGESTERONE ACETATE, DROSPIRENONE AND NESTORONE ON BREAST CANCER CELL MIGRATION AND INVASION”.

We have made the final changes as requested by the reviewers and changed the competing interest statement as requested by the Editor.

The MS complies with all the BMC Cancer editorial rules, and we thus hope that it will be now fit for publication on the BMC Cancer.

Please, be assured that the manuscript has not been published nor is being considered for publication elsewhere in whole or in part, in any language.

Yours Sincerely,

Tommaso Simoncini, MD, PhD,
(on behalf of the authors)

Molecular and Cellular Gynecological Endocrinology Laboratory
Department of Reproductive Medicine and Child Development
Division of Obstetrics and Gynecology
University of Pisa

Via Roma, 67, 56100
PISA, ITALY

Tel: +39.050.992690
Fax: +39.050.553410
E-mail: t.simoncini@obgyn.med.unipi.it
Web: http://www.med.unipi.it/MCGEL

Pisa. March 22, 2008